Author's response to reviews

Title: Case Report: A Rare Lesion Around The Knee Synovial Hemangiohamartomas

Authors:

Serkan Senol (drserkansenol@yahoo.com)
Hakan Cift (hakanturancift@yahoo.com)
Korhan Ozkan (korhanozkan@hotmail.com)
Esat Uygur (esatuygur@hotmail.com)
Maria Silvia Spinelli (msilvia@yahoo.it)

Version: 2 Date: 17 February 2012

Author's response to reviews:

Unexpected or unusual presentations of a disease.